Author's response to reviews

Title: Cluster Randomisation or Randomised Consent as an Appropriate Methodology for Trials in Palliative Care: a Feasibility Study ISRCTN60243484

Authors:

Andrew A Fowell Dr (andrew.fowell@nww-tr.wales.nhs.uk)
Ian I Russell Professor (ian.russell@bangor.ac.uk)
Mrs Ros R Johnstone Mrs (wccpp.manager@nww-tr.wales.nhs.uk)
Ilora I Finlay Professor (ilora.finlay@velindre.tr.wales.nhs.uk)
Daphne D Russell Dr (d.russell@bangor.ac.uk)

Version: 2 Date: 8 Apr 2004

To whom it may concern, revisions to the manuscript 7822655573154709 concern formatting of the title page and minor editing within the text to clearly distinguish between zelen method and cluster randomisation. Yours sincerely Ros Johnstone
(Author for correspondence)